Author’s response to reviews

Title: Frequency of foot ulcers in people with type 2 diabetes, presenting to specialist diabetes clinic at a tertiary care hospital, Lahore, Pakistan”.

Authors:

Bilal Younus (bilalbin@gmail.com)
Adeela Shahid (dradeela71@gmail.com)
Rozina Arshad (rozina979@gmail.com)
Saima Khursheed (dr.cyma@yahoo.com)
Muhammad Ahmad (mlashari7@gmail.com)
Haroon Yousaf (doctharoon@gmail.com)

Version: 1 Date: 20 Jun 2018

Author’s response to reviews:

Copy attached separately